World History 2nd Semester Final Exam Review Sheet

What is historically significant about these terms;

Huguenots	divine right	mercenary	Segregation	abolition movement
electorate	Louisiana Purchase	Trade balance	trade surplus	trade deficit
mercantilism	Jamestown	Monroe Doctrine	Hermit Kingdom	French Indochina
total war	stalemate	neutrality	Bolsheviks	Central Powers
republic	scorched earth	social darwinism	reparations	propaganda
privateers	guillotine	gulag	apartheid	flappers
Gestapo	totalitarian state	Nuremberg Laws	nationalism	Nazism
Facism	Hiroshima	appeasement	Nuremberg Trials	blitzkrieg
anti-semitism	Rosie the Riveter	Manhattan Project	Kamikaze	D-Day
Communism	Truman Doctrine			

Who led the first successful mission to circumnavigate the globe? Ferdinand Magellan

How did European expansion of the slave trade affect African states in the 1600s and 1700s? African states began to compete and kidnap enemies to dominate the slave trade.

What did the islands that Columbus discovered on his voyage become known as?

The West Indies

How did Europeans usually acquire African captives for the slave trade? African traders brought captives to ports.

What was significant about the settlement at Cape Town, South Africa? First permanent European settlement in Africa

To conquer the Aztec empire, Hernán Cortés formed alliances with conquered people who hated the Aztecs.

This conquistador conquered present day Peru, Ecuador and Chile. Pizarro

The English colonies of Massachusetts, Pennsylvania, and Maryland were mainly set up as havens for persecuted religious groups.

When he encountered the Taino people in the West Indies, Christopher Columbus claimed their land for the Spanish king.

What was the basic ideology of The Enlightenment? Science and intellectual explanations as opposed to superstition, intolerance and abuses in church and state.

2nd Semester Final Exam Review History

Ensheng Dong

 ${\bf 2nd} \; {\bf Semester} \; {\bf Final} \; {\bf Exam} \; {\bf Review} \; {\bf History:}$

Unveiling the Energy of Verbal Beauty: An Emotional Sojourn through 2nd Semester Final Exam Review History

In some sort of inundated with monitors and the cacophony of immediate communication, the profound energy and psychological resonance of verbal beauty usually diminish in to obscurity, eclipsed by the regular barrage of sound and distractions. Yet, nestled within the lyrical pages of **2nd Semester Final Exam Review History**, a charming function of fictional beauty that pulses with raw feelings, lies an unique trip waiting to be embarked upon. Written by a virtuoso wordsmith, this mesmerizing opus guides readers on a psychological odyssey, gently revealing the latent possible and profound influence stuck within the elaborate web of language. Within the heart-wrenching expanse of the evocative analysis, we can embark upon an introspective exploration of the book is key themes, dissect its interesting publishing type, and immerse ourselves in the indelible impact it leaves upon the depths of readers souls.

http://www.a-walhalla.hu/results/virtual-library/fetch.php/advanced%20pressure%20points%20fighting.pdf

Table of Contents 2nd Semester Final Exam Review History

- 1. Understanding the eBook 2nd Semester Final Exam Review History
 - The Rise of Digital Reading 2nd Semester Final Exam Review History
 - Advantages of eBooks Over Traditional Books
- 2. Identifying 2nd Semester Final Exam Review History
 - Exploring Different Genres
 - Considering Fiction vs. Non-Fiction
 - Determining Your Reading Goals
- 3. Choosing the Right eBook Platform
 - Popular eBook Platforms
 - Features to Look for in an 2nd Semester Final Exam Review History
 - User-Friendly Interface
- 4. Exploring eBook Recommendations from 2nd Semester Final Exam Review History
 - Personalized Recommendations

- 2nd Semester Final Exam Review History User Reviews and Ratings
- 2nd Semester Final Exam Review History and Bestseller Lists
- 5. Accessing 2nd Semester Final Exam Review History Free and Paid eBooks
 - 2nd Semester Final Exam Review History Public Domain eBooks
 - 2nd Semester Final Exam Review History eBook Subscription Services
 - o 2nd Semester Final Exam Review History Budget-Friendly Options
- 6. Navigating 2nd Semester Final Exam Review History eBook Formats
 - o ePub, PDF, MOBI, and More
 - 2nd Semester Final Exam Review History Compatibility with Devices
 - 2nd Semester Final Exam Review History Enhanced eBook Features
- 7. Enhancing Your Reading Experience
 - Adjustable Fonts and Text Sizes of 2nd Semester Final Exam Review History
 - Highlighting and Note-Taking 2nd Semester Final Exam Review History
 - Interactive Elements 2nd Semester Final Exam Review History
- 8. Staying Engaged with 2nd Semester Final Exam Review History
 - Joining Online Reading Communities
 - Participating in Virtual Book Clubs
 - Following Authors and Publishers 2nd Semester Final Exam Review History
- 9. Balancing eBooks and Physical Books 2nd Semester Final Exam Review History
 - Benefits of a Digital Library
 - Creating a Diverse Reading Collection 2nd Semester Final Exam Review History
- 10. Overcoming Reading Challenges
 - Dealing with Digital Eye Strain
 - Minimizing Distractions
 - Managing Screen Time
- 11. Cultivating a Reading Routine 2nd Semester Final Exam Review History
 - Setting Reading Goals 2nd Semester Final Exam Review History
 - Carving Out Dedicated Reading Time
- 12. Sourcing Reliable Information of 2nd Semester Final Exam Review History
 - Fact-Checking eBook Content of 2nd Semester Final Exam Review History

- Distinguishing Credible Sources
- 13. Promoting Lifelong Learning
 - Utilizing eBooks for Skill Development
 - Exploring Educational eBooks
- 14. Embracing eBook Trends
 - Integration of Multimedia Elements
 - Interactive and Gamified eBooks

2nd Semester Final Exam Review History Introduction

In this digital age, the convenience of accessing information at our fingertips has become a necessity. Whether its research papers, eBooks, or user manuals, PDF files have become the preferred format for sharing and reading documents. However, the cost associated with purchasing PDF files can sometimes be a barrier for many individuals and organizations. Thankfully, there are numerous websites and platforms that allow users to download free PDF files legally. In this article, we will explore some of the best platforms to download free PDFs. One of the most popular platforms to download free PDF files is Project Gutenberg. This online library offers over 60,000 free eBooks that are in the public domain. From classic literature to historical documents, Project Gutenberg provides a wide range of PDF files that can be downloaded and enjoyed on various devices. The website is user-friendly and allows users to search for specific titles or browse through different categories. Another reliable platform for downloading 2nd Semester Final Exam Review History free PDF files is Open Library. With its vast collection of over 1 million eBooks, Open Library has something for every reader. The website offers a seamless experience by providing options to borrow or download PDF files. Users simply need to create a free account to access this treasure trove of knowledge. Open Library also allows users to contribute by uploading and sharing their own PDF files, making it a collaborative platform for book enthusiasts. For those interested in academic resources, there are websites dedicated to providing free PDFs of research papers and scientific articles. One such website is Academia.edu, which allows researchers and scholars to share their work with a global audience. Users can download PDF files of research papers, theses, and dissertations covering a wide range of subjects. Academia.edu also provides a platform for discussions and networking within the academic community. When it comes to downloading 2nd Semester Final Exam Review History free PDF files of magazines, brochures, and catalogs, Issuu is a popular choice. This digital publishing platform hosts a vast collection of publications from around the world. Users can search for specific titles or explore various categories and genres. Issuu offers a seamless reading experience with its user-friendly interface and allows users to download PDF files for offline reading. Apart from dedicated platforms, search engines also play a crucial role in finding free PDF files. Google, for

instance, has an advanced search feature that allows users to filter results by file type. By specifying the file type as "PDF," users can find websites that offer free PDF downloads on a specific topic. While downloading 2nd Semester Final Exam Review History free PDF files is convenient, its important to note that copyright laws must be respected. Always ensure that the PDF files you download are legally available for free. Many authors and publishers voluntarily provide free PDF versions of their work, but its essential to be cautious and verify the authenticity of the source before downloading 2nd Semester Final Exam Review History. In conclusion, the internet offers numerous platforms and websites that allow users to download free PDF files legally. Whether its classic literature, research papers, or magazines, there is something for everyone. The platforms mentioned in this article, such as Project Gutenberg, Open Library, Academia.edu, and Issuu, provide access to a vast collection of PDF files. However, users should always be cautious and verify the legality of the source before downloading 2nd Semester Final Exam Review History any PDF files. With these platforms, the world of PDF downloads is just a click away.

FAQs About 2nd Semester Final Exam Review History Books

What is a 2nd Semester Final Exam Review History PDF? A PDF (Portable Document Format) is a file format developed by Adobe that preserves the layout and formatting of a document, regardless of the software, hardware, or operating system used to view or print it. How do I create a 2nd Semester Final Exam Review History PDF? There are several ways to create a PDF: Use software like Adobe Acrobat, Microsoft Word, or Google Docs, which often have built-in PDF creation tools. Print to PDF: Many applications and operating systems have a "Print to PDF" option that allows you to save a document as a PDF file instead of printing it on paper. Online converters: There are various online tools that can convert different file types to PDF. How do I edit a 2nd Semester Final Exam Review History PDF? Editing a PDF can be done with software like Adobe Acrobat, which allows direct editing of text, images, and other elements within the PDF. Some free tools, like PDFescape or Smallpdf, also offer basic editing capabilities. How do I convert a 2nd Semester Final Exam Review **History PDF to another file format?** There are multiple ways to convert a PDF to another format: Use online converters like Smallpdf, Zamzar, or Adobe Acrobats export feature to convert PDFs to formats like Word, Excel, JPEG, etc. Software like Adobe Acrobat, Microsoft Word, or other PDF editors may have options to export or save PDFs in different formats. How do I password-protect a 2nd Semester Final Exam Review History PDF? Most PDF editing software allows you to add password protection. In Adobe Acrobat, for instance, you can go to "File" -> "Properties" -> "Security" to set a password to restrict access or editing capabilities. Are there any free alternatives to Adobe Acrobat for working with PDFs? Yes, there are many free alternatives for working with PDFs, such as: LibreOffice: Offers PDF editing features. PDFsam: Allows splitting,

merging, and editing PDFs. Foxit Reader: Provides basic PDF viewing and editing capabilities. How do I compress a PDF file? You can use online tools like Smallpdf, ILovePDF, or desktop software like Adobe Acrobat to compress PDF files without significant quality loss. Compression reduces the file size, making it easier to share and download. Can I fill out forms in a PDF file? Yes, most PDF viewers/editors like Adobe Acrobat, Preview (on Mac), or various online tools allow you to fill out forms in PDF files by selecting text fields and entering information. Are there any restrictions when working with PDFs? Some PDFs might have restrictions set by their creator, such as password protection, editing restrictions, or print restrictions. Breaking these restrictions might require specific software or tools, which may or may not be legal depending on the circumstances and local laws.

Find 2nd Semester Final Exam Review History:

advanced pressure points fighting stand tall molly lou melon

practice 8 2 special right triangles answers

john von neumann and the foundations of quantum physics

topcon gts 300 user manual

link belt ls 2800 operators manual

manual bombardier gsx

volvo penta shop manual a125 ohio right to life voter quide

wiring for 1993 f250 speedo

manual bolens bl100

read muggie maggie unit plan

6 hp evinrude outboard repair manual

the banished princess

campbell green bean casserole receipe

2nd Semester Final Exam Review History:

free free aboriginal templates for writing 3 4 twinkl - Jul 23 2023

web how do the free aboriginal templates for writing work use this differentiated resource to plan a story in the style of an

aboriginal dreaming story show more related searches dream time stories story plan story writing aboriginal culture narrative writing how the birds got their colours ratings reviews curriculum links make a request aboriginal dreaming stories resource pack teacher made twinkl - Jan 17 2023

web introduce your children to the stories of the dreaming with this aboriginal dreaming stories for kids resources pack this resource pack contains five of the most well known and beloved dreaming stories including the rainbow serpent tiddalick the frog and how the birds got their colours

aboriginal dreamtime stories japingka aboriginal art gallery - Dec 16 2022

web the dreamtime is a term that describes unique stories and beliefs owned and held by different australian aboriginal groups the history of the dreamtime word and its meanings says something about the development of the ideas held about the aboriginal world and how they are expressed through art

how to write a dreamtime story pen and the pad - Jun 22 2023

web how to write a dreamtime story the indigenous people of australia wrote dreamtime stories to explain how the familiar world came into existence dreamtime or dreaming refers to the time of creation when the great spirit ancestors traveled across the australian continent forming the land animals and even the

18 858 templates stock photos images pictures dreamstime - May 21 2023

web download templates stock photos free or royalty free photos and images use them in commercial designs under lifetime perpetual worldwide rights dreamstime is the world's largest stock photography community

1 161 top write your own dreamtime story teaching - Feb 06 2022

web 1 161 top write your own dreamtime story teaching resources curated for you the rainbow serpent book review writing worksheet 4 9 7 reviews tiddalick the frog book review writing template why the koala has a stumpy tail book review writing worksheet 4 5 4 reviews how to write an australian aboriginal inspired dreaming story

aboriginal dreamtime stories summary culture examples - Sep 13 2022

web may 16 2022 instructor kerry gray view bio learn about the aboriginal dreamtime stories and art understand the deep connection of dreamtime creation stories with aboriginal culture through relevant

dreamtime story template secure4 khronos - Apr 08 2022

web mystified to enjoy every book selections dreamtime story template that we will absolutely offer along with handbooks you could take pleasure in the present is dreamtime story

5 popular dreamtime stories for kids children s ground - Mar 19 2023

web 5 popular dreamtime stories for kids hello there storytelling aficionados gather around for a fantastic journey into the heart of australia today well be exploring the vibrant and mystical world of aboriginal dreamtime stories

how to write a dreamtime story for kids best writing service - Mar 07 2022

web professional authors can write an essay in 3 hours if there is a certain volume but it must be borne in mind that with such a service the price will be the highest the cheapest estimate is the work that needs to be done in 14 days then 275 words will cost you 10 while 3 hours will cost you 50 please take into consideration that vat

free free aboriginal templates for writing 3 4 twinkl - May 09 2022

web this set of free aboriginal templates for writing will help children get inspired by dreaming stories and create one of their very own each template will guide children through the creative writing process and will fit well in lessons teaching children about aboriginal culture show more related searches

aboriginal dreamtime stories dreamtime - Aug 24 2023

web dreamtime is a portal that allows any person from any race religion gender and generation to step into the world of the aboriginal dreamtime please take your time to explore the different stories on offer and share your thoughts and feelings the dreaming stories and songlines teach starter - Nov 15 2022

web a 60 minute lesson in which students will identify the importance of the dreaming dreamtime stories and songlines in aboriginal cultures login to view the lesson plan curriculum australian curriculum v9 australian curriculum v8 new south wales curriculum victorian curriculum ac9hs3k04

dreamtime stories teach starter - Apr 20 2023

web results for dreamtime stories 413 teaching resources dreamtime stories sort relevance year levels foundation year 138 preschool kindergarten 28 year 1 235

126 top dreamtime stories teaching resources curated for you twinkl - Jan 05 2022

web why the koala has a stumpy tail storyboard template 5 0 2 reviews aboriginal dreaming wuriunpranilli the sun woman story cards 4 7 9 reviews explore more than 126 dreamtime stories resources for teachers parents and pupils as well as related resources on naidoc week 2023

free tiddalick the frog stick puppets aboriginal dreamtime - Oct 14 2022

web free tiddalick the frog stick puppets aboriginal dreamtime create outcome 1 identity and family outcome 2 belonging and community f 2 aboriginal dreaming stories respond to read and view texts free resource download now free download free tiddalick the frog stick puppets 4 9 34 reviews

dreamtime story template secure4 khronos - Jul 11 2022

web jun 3 2023 every ideal location within online connections you could buy guide dreamtime story template or get it as soon as possible acknowledging the exaggeration ways to acquire this ebook dreamtime story template is moreover beneficial by looking the title publisher or authors of instruction you in truly want you can discover them swiftly

results for writing dreamtime stories tpt - Feb 18 2023

web the templates can be adapted to suit a range of activities including retell of the story lesson learnt from the story the origin of the story and more it also includes a drawing of the bird which can be used as a craftivity

explore a dreaming story worksheet teach starter - Jun 10 2022

web 1 2 available on the plus plan use this template to explore first nations dreaming stories and unlock how they connect to country explore first nations dreaming stories storytelling is a sacred tradition shared by the first nations people of australia encourage your students to explore a dreaming story that comes from your area s country

how to write a dreamtime story for kids top writers - Aug 12 2022

web jan 3 2021 how to write a dreamtime story for kids research papers obsity in children level college university high school master s phd undergraduate regular writer

between us 2022 full cast crew mydramalist - Aug 21 2023

web drama between us country thailand episodes 12 aired nov 6 2022 jan 29 2023 aired on sunday original network gmm one iqiyi duration 53 min genres romance tags bisexual male lead friends with benefits lgbtq swimming gay romance trauma friendship post traumatic stress disorder ptsd gay male lead childhood

between us 2022 ep 1 eng sub video dailymotion - May 18 2023

betweenus - Jul 20 2023

web about us betweenus in is a parent teacher portal communication for podar group of schools pan india it is owned by podar education network which runs podar group of schools in the country

between us wikipedia - Feb 15 2023

web between us between us may refer to between us 2003 film a short film by laurits munch petersen between us 2004 film a short film by charlotte bruus christensen between us 2011 film a mexican comedy film between us 2012 film an american film based on the play of the same name between us 2016 film an american drama film

between us 2022 episodes mydramalist - Apr 17 2023

web nov 6 2022 drama between us country thailand episodes 12 aired nov 6 2022 jan 29 2023 aired on sunday original network gmm one iqiyi duration 53 min genres romance tags bisexual male lead friends with benefits lgbtq swimming gay romance trauma friendship post traumatic stress disorder ptsd gay male

between us 2022 full online with english subtitle for free iqiyi - Mar 16 2023

web watch the latest thai drama thai lagoon between us 2022 full online with english subtitle for free on iqiyi iq com team

always has a sleeping issue before the swimming competition which makes him unable to make good between us drama wiki fandom - Jan 14 2023

web directed by new siwaj sawatmaneekul starring boun noppanut guntachai as win prem warut chawalitrujiwong as team country of origin thailand original language thai no of episodes 16 timeslot sunday 10 45 gmt 7 episode runtime between us 2021 mydramalist - Jun 19 2023

web mar 6 2021 buy on amazon add to list 6 7 your rating 0 10 ratings 6 7 10 from 253 users of watchers 1 926 reviews 2 users two love stories centered around students attending the same university sai and khel meet at the school library while trying to check out the same book

between best app for couples - Sep 22 2023

web between is a mobile app for couples in love chat track anniversaries share photos and video and plan your schedules together all in one private space start using between now and make your relationship even better

between us 2022 mydramalist - Oct 23 2023

web nov 6 2022 ratings 7 9 10 from 16 266 users of watchers 40 719 reviews 136 users team is a talented swimmer who just entered university however when it comes to swim competitions he s never able to perform to the best of his abilities the rise and fall of the trigan empire volume iv paperback - Mar 18 2022

web genre super heroes buy new 18 99 rrp 23 99 details save 5 00 21 free returns free delivery friday may 5 details or fastest delivery wednesday may 3 order within 15 hrs 56 mins details select delivery location in stock quantity add to basket buy now payment secure transaction dispatches from amazon sold by

the rise and fall of the trigan empire volume i volume 1 - Sep 04 2023

web 54 24 free delivery on first order details sold by amazon au roll over image to zoom in read sample follow the author don lawrence the rise and fall of the trigan empire volume 1 paperback 19 march 2020 by

the rise and fall of the trigan empire volume i volume 1 - Jan 28 2023

web mar 17 2020 12 91 available instantly paperback 33 51 other used new collectible from 27 24 buy new 33 51 free delivery thursday august 24 on your first order details or fastest delivery tomorrow august 21 order within 12 hrs 48 mins details select delivery location in stock quantity add to cart buy now secure transaction

the rise and fall of the trigan empire volume v amazon com - Jun 20 2022

web 32 95 free returns free delivery october 27 31 or fastest delivery wednesday october 25 select delivery location this title will be released on october 24 2023 qty 1 pre order now payment secure transaction ships from amazon com sold by amazon com returns eligible for return refund or replacement within 30 days of

the rise and fall of the trigan empire volume one the runaway 1 - Oct 05 2023

web buy the rise and fall of the trigan empire volume one the runaway 1 bestselling graphic novel volume 1 1 by don lawrence isbn 9781781087558 from amazon s book store everyday low prices and free delivery on eligible orders

the rise and fall of the trigan empire volume i volume 1 - Aug 23 2022

web the rise and fall of the trigan empire volume i volume 1 lawrence don amazon sg books

the rise and fall of the trigan empire volume ii volume 2 - Jul 22 2022

web 43 01 17 54 delivery sold by amazon us roll over image to zoom in read sample follow the author don lawrence the rise and fall of the trigan empire volume ii volume 2 paperback 24 december 2020 by don lawrence author mike butterworth author 4 8 290 ratings part of the rise and fall of the trigan empire

the rise and fall of the trigan empire volume one amazon com - Mar 30 2023

web mar 19 2020 the rise and fall of the trigan empire volume one kindle edition by butterworth mike lawrence don download it once and read it on your kindle device pc phones or tablets use features like bookmarks note taking and highlighting while reading the rise and fall of the trigan empire volume one

the rise and fall of the trigan empire volume iii volume 3 - Nov 25 2022

web 49 69 free delivery sold by pbshopuk au tracked roll over image to zoom in read sample follow the author don lawrence the rise and fall of the trigan empire volume iii volume 3 paperback 20 july 2021 by don lawrence author 4 8 215 ratings part of the rise and fall of the trigan empire see all formats and editions

the rise and fall of the trigan empire volume iii 3 - Sep 23 2022

web oct 5 2021 this is the epic story of the trigan empire s rise and fall and of how trigo often alone had to fend off usurpers and monstrous threats to save his people the lush painted comic art that don lawrence produced in this period would solidify him as one of the greatest comic book artists of all time

in review the rise and fall of the trigan empire volume three - Apr 30 2023

web jul 26 2021 by peter duncan on july 26 2021 2 peter duncan reviews the latest volume in the epic story of the trigan empire s rise and fall and also takes a look at artist don lawrence s follow up project storm by mike butterworth and don lawrence additional strip by miguel quesada

the rise and fall of the trigan empire volume i waterstones - May 20 2022

web mar 19 2020 the first of a four volume series reprinting the rise and fall of the trigan empire a landmark 1960s science fiction series which rivalled game of thrones in popularity and was the precursor to every mythic sci fi adventure to come the rise and fall of the trigan empire 1 volume i issue - Feb 26 2023

web mar 18 2020 co created by mike butterworth and don lawrence the rise and fall of the trigan empire has been named as a major influence by the likes of writer neil gaiman sandman artist dave gibbons

the rise and fall of the trigan empire volume three volume 3 - Oct 25 2022

web jul 22 2021 details select delivery location in stock quantity add to basket buy now payment secure transaction dispatches from amazon sold by amazon returns returnable within 30 days of receipt add gift options buy used 12 08 new used 18 from 1208 2 80 delivery have one to sell sell on amazon

the rise and fall of the trigan empire volume iii goodreads - Dec 27 2022

web the best selling treasury of british comics archival series the rise and fall of the trigan empire reaches its third thrilling volume the third thrilling omnibus of the lost sci fi classic from the sixties that the new york times noted had the trigan empire wikipedia - Aug 03 2023

web the rise and fall of the trigan empire rebellion volume i 304 pages march 2020 isbn 978 1 78108 755 8 comprises 13 stories published from september 18th 1965 to may 18th 1968 volume ii 288 pages december 2020 isbn 978 1 78108 775 6 comprises 11 stories published from may 25th 1968 to november 14th 1970

the rise and fall of the trigan empire volume 1 rebellion - Feb 14 2022

web may 5 2023 originally published in the anthology titles ranger from the first issue in september 1965 and look and learn from june 1966 until april 1982 the series told the story of an alien empire on the planet elekton that was heavily influenced by history particularly roman

the rise and fall of the trigan empire volume i goodreads - Jun 01 2023

web apr $14\ 2020$ the first of a four volume series reprinting the rise and fall of the trigan empire a landmark 1960s science fiction series which rivalled game of thrones in popularity and was the precursor to every mythic sci fi adventure to come the rise and fall of the trigan empire volume i 1 - Jul $02\ 2023$

web mar 17 2020 the first of a four volume series reprinting the rise and fall of the trigan empire a landmark 1960s science fiction series which rivalled game of thrones in popularity and was the precursor to every mythic sci fi adventure to come the rise and fall of the trigan empire volume v paperback - Apr 18 2022

web this fifth omnibus collection of the science fiction classic the rise and fall of the trigan empire reprints all the stories originally published in look learn from 1975 through to 1977 and contains don lawrence s final flawlessly painted trigan empire pages